Author's response to reviews

Title: Chronic disease prevalence and care among elderly in urban and rural Beijing, China - a 10/66 Dementia Research Group cross-sectional survey

Authors:

Zhaorui Liu (zhaoruiliu@gmail.com)
Emiliano Albanese (emiliano.albanese@iop.kcl.ac.uk)
Shuran Li (lishuran@bjmu.edu.cn)
Yueqin Huang (dengy@mail.tsinghua.edu.cn)
Cleusa P Ferri (cleusa.ferri@iop.kcl.ac.uk)
Fang Yan (emillyyf@163.com)
Renata Sousa (renata.sousa@iop.kcl.ac.uk)
Weimin Dang (dangweimin@bjmu.edu.cn)
Martin Prince (martin.prince@iop.kcl.ac.uk)

Version: 5 Date: 1 October 2009

Author's response to reviews: see over
Dear BMC Public Health Associate Editor,

Thank you for your email. We appreciate your helpful additional comments and suggestions and have revised our paper accordingly. Our responses and changes are summarised below:

**Major revisions**

(1) Reference 1, 2, 8, 15, 43, 46 - please provide full details for these references. For an Article we require the year/volume/issue/page numbers and for a book the publisher/city/year. If an article is in press please indicate this in the reference list. Alternatively please provide an open access weblink

Done.

(2) Reference 4, 48 incomplete - Please provide volume and page numbers or article number. If the article is in press please indicate this in the reference list

Done.

(3) Reference 18 - Please provide the publisher's details

Done.

**Minor revisions**

(1) Title - Please remove this text from above the title

Done.

(2) Authors - Please remove this text from above the author list

Done.

(3) Word count/ key words - Please remove these

Done.

(4) 17/09/2009 - Please remove this footer

Done.

(5) Acknowledgments - Please move this section to after the authors¿ contributions

Done.

(6) Authors¿ contributions - Please use initials to refer to each author

Done.

(7) Tables - Please move these to a position after the reference list

Done.
Thank you again for all the comments and suggestion. If you think we need add more information, please let us know.

Best Regards,

Dr. Zhaorui Liu; Dr. Emiliano Albanese; Prof. Yueqin Huang and Professor Martin Prince for and on behalf of the 10/66 Dementia Research Group